Author's response to reviews

Title: Type 3 finger length pattern is associated with total knee replacements due to osteoarthritis but not with hip replacements or hand osteoarthritis in the elderly: The AGES-Reykjavik Study.

Authors:

Kristin Sigurjonsdottir (krs2@hi.is)
Thorir M Bjorgulfsson (thb1@hi.is)
Thor Aspelund (asgelund@hjarta.is)
Gudny Eiriksdottir (gudny@hjarta.is)
Sigurdur Sigurdsson (sigurdur@hjarta.is)
Tamara B Harris (harrista@nia.nih.gov)
Lenore Launer (launrl@nia.nih.gov)
Vilmundur Gudnason (v.gudnason@hjarta.is)
Helgi Jonsson (helgijon@landspitali.is)

Version: 2 Date: 18 October 2012

Author's response to reviews:

In the amended version I have added the following sentence in the methods section:

"The study is approved by the Icelandic National Bioethics Committee, (VSN: 00-063) and the Data Protection Authority."